Author's response to reviews

Title: ENMD-1198, a novel tubulin-binding agent reduces HIF-1 and STAT3 activity in human hepatocellular carcinoma (HCC) cells, and inhibits growth and vascularization in vivo

Authors:

Christian Moser (christian.moser@klinik.uni-regensburg.de)
Sven A Lang (sven.lang@klinik.uni-regensburg.de)
Akira Mori (akr@kuhp.kyoto-u.ac.jp)
Claus Hellerbrand (claus.hellerbrand@klinik.uni-regensburg.de)
Hans J Schlitt (hans.schlitt@klinik.uni-regensburg.de)
Edward K Geissler (edward.geissler@klinik.uni-regensburg.de)
William E Fogler (fogler@entremed.com)
Oliver Stoeltzing (oliver.stoeltzing@klinik.uni-regensburg.de)

Version: 7 Date: 16 July 2008

Author's response to reviews:

The requested formatting changes have been made. Thank you very much for your final decision letter and comments.

Kind regards

Oliver Stoeltzing